Author’s response to reviews

Title: A Gaze-independent Audiovisual Brain-Computer Interface for Detecting Awareness of Patients with Disorders of Consciousness

Authors:
Jiahui Pan (panjh82@qq.com)
Jiahui Pan (panjh82@qq.com)
Yan Chen (269837537@qq.com)
Yanbin He (17336071@qq.com)
Xiaoxiao Ni (862121021@qq.com)
Jiechun Zhang (353336293@qq.com)
Fei Wang (sandra88@qq.com)
Yuanqing Li (auyqli@scut.edu.cn)
Ronghao Yu (gesund@21cn.com)

Version: 2 Date: 12 Apr 2018

Author’s response to reviews:

Dear Editor,

The authors are highly grateful to the editors for coordinating the review of our manuscript, and to the reviewers for the insightful and constructive comments! By revising the manuscript strenuously based on your comment, the authors believe that the paper has been enhanced substantially. The point-to-point responses to your comments are listed in the attach files.